Author’s response to reviews

Title: Assessing social recovery of vulnerable youth in global mental health settings: A pilot study of clinical research tools in Malaysia.

Authors:

Clio Berry (c.berry@sussex.ac.uk)
Ellisha Othman (ellisha@sols247.org)
Jun Chen Tan (jun@sols247.org)
Brioney Gee (brioney.gee@nsft.nhs.uk)
Rory Edward Byrne (rory.byrne@gmmh.nhs.uk)
Joanne Hodgekins (j.hodgekins@uea.ac.uk)
Daniel Michelson (D.Michelson@sussex.ac.uk)
Alvin Lai Oon Ng (alvinn@sunway.edu.my)
Nigel V Marsh (nigelm@sunway.edu.my)
Sian Coker (s.coker@uea.ac.uk)
David Fowler (d.fowler@sussex.ac.uk)

Version: 1 Date: 03 Dec 2018

Author’s response to reviews:

1. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please clarify whether the consent for publication had been collected for the explicit purpose of the publication of potentially identifiable information. If not then please see request number 2.

2. We note that the dataset for table 1 contains potentially identifiable information, i.e. age, sex, etc. Please amend the dataset to address this issue, by removing the age and sex columns of data.
Alternatively, you may choose to remove this table altogether as your Availability of data and materials statement already indicates that the datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

In either case, please also amend your Consent for publication statement so that it appears as "Not applicable" as no personal information would be provided in your manuscript.

Participants consented to the use of information they provided in publications on the basis that it was anonymous. As it seems that BMC Psychiatry treat age and sex as potentially identifiable, Table 1 has been removed and the consent statement amended. Aggregate age and sex is provided in text but is aggregate data only thus we do not consider this in anyway personally identifying.

3. Please include the email addresses of all authors on the title page of your manuscript.

I confirm all email addresses have been provided.

4. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

I confirm the list of abbreviations has been provided and text checked.

5. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

I confirm such a manuscript has been submitted.